Author’s response to reviews

Title: Therapeutic plasma exchange as rescue therapy in severe sepsis and septic shock: retrospective observational single-center study of 23 patients

Authors:

Johannes Hadem (hadem.johannes@mh-hannover.de)
Carsten Hafer (c.hafer@comlink.org)
Andrea S. Schneider (andrea.s.schneider@gmx.de)
Olaf Wiesner (wiesner.olaf@mh-hannover.de)
Gernot Beutel (beutel.gernot@mh-hannover.de)
Thomas Fuehner (fuehner.thomas@mh-hannover.de)
Tobias Welte (welte.tobias@mh-hannover.de)
Marius M. Hoeper (hoeper.marius@mh-hannover.de)
Jan T. Kielstein (kielstein@yahoo.com)

Version: 2 Date: 21 April 2013

Author’s response to reviews:

I have removed our point-by-point response associated with an earlier submission as you suggested.

If the reviewers feel, any potential identifier should be removed from one of our tables, we would of course be ready to modify this.